

## G OPEN ACCESS

**Citation:** Eales O, Shearer FM, McCaw JM (2025) How immunity shapes the long-term dynamics of influenza H3N2. PLoS Comput Biol 21(3): e1012893. <u>https://doi.org/10.1371/</u> journal.pcbi.1012893

**Editor:** Eric HY Lau, The University of Hong Kong, CHINA

Received: March 13, 2024

Accepted: February 21, 2025

Published: March 20, 2025

**Copyright:** © 2025 Eales et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data availability statement: Code supporting this paper is available at <u>https://github.com/</u> <u>Eales96/flu-long-term-agent</u> (Appache-2.0 license). Reused publicly available data (antigenic map used to fit antigenic drift model) is available at <u>https://elifesciences.org/</u> <u>articles/01914/figures#SD1-data</u> (Bedford et al. 2014 <u>https://doi.org/10.7554/eLife.01914</u>).

**Funding:** OE is supported by a University of Melbourne McKenzie Fellowship. FMS is

RESEARCH ARTICLE

# How immunity shapes the long-term dynamics of influenza H3N2

#### Oliver Eales 1,2\*, Freya M. Shearer<sup>1,3‡</sup>, James M. McCaw<sup>1,2‡</sup>

1 Infectious Disease Dynamics Unit, Centre for Epidemiology and Biostatistics, Melbourne School of Population and Global Health, The University of Melbourne, Melbourne, Australia, 2 School of Mathematics and Statistics, The University of Melbourne, Melbourne, Australia, 3 Infectious Disease Ecology and Modelling, The Kids Research Institute, Perth, Australia

‡ These authors are joint senior authors on this work.
\* oliver.eales@unimelb.edu.au

# Abstract

Since its emergence in 1968, influenza A H3N2 has caused yearly epidemics in temperate regions. While infection confers immunity against antigenically similar strains, new antigenically distinct strains that evade existing immunity regularly emerge ('antigenic driff'). Immunity at the individual level is complex, depending on an individual's lifetime infection history. An individual's first infection with influenza typically elicits the greatest response with subsequent infections eliciting progressively reduced responses ('antigenic seniority'). The combined effect of individual-level immune responses and antigenic drift on the epidemiological dynamics of influenza are not well understood. Here we develop an integrated modelling framework of influenza transmission, immunity, and antigenic drift to show how individual-level exposure, and the build-up of population level immunity, shape the long-term epidemiological dynamics of H3N2. Including antigenic seniority in the model, we observe that following an initial decline after the pandemic year, the average annual attack rate increases over the next 80 years, before reaching an equilibrium, with greater increases in older age-groups. Our analyses suggest that the average attack rate of H3N2 is still in a growth phase. Further increases, particularly in the elderly, may be expected in coming decades, driving an increase in healthcare demand due to H3N2 infections.

### Author summary

Influenza A H3N2 causes yearly epidemics in temperate regions causing substantial mortality and morbidity, particularly in older individuals. Infection with influenza elicits an immune response, and antibodies are produced that protect against future infection with similar strains of the virus. However, due to the gradual accumulation of mutations (evolution) novel strains of influenza regularly emerge that are not well-recognised by an individual's antibodies, and so can evade an individual's existing immunity causing repeat infections. An individual's overall immunity against a particular strain is complex and depends on their entire infection history. An individual's first infection typically produces the greatest antibody response, with future infections producing progressively reduced

supported by the National Health and Medical Research Council of Australia through the Investigator Grant Scheme (Emerging Leader Fellowship, 2021/GNT2010051). JMM is supported by the Australian Research Council through the Laureate Fellowship Scheme (FL240100126). This research is supported by the Australian Consortium of Epidemic Forecasting and Analytics (ACEFA), a National Health and Medical Research Council of Australia Centre of Research Excellence (2035303). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

responses. This is known as antigenic seniority. We use a mathematical modelling framework to explore the role of individual-level immunity and repeat infections in shaping long-term population-level influenza dynamics. We find that when antigenic seniority is included in the model of individual-level immunity, that the average annual attack rate of influenza is expected to increase, following a minimum in the years after a pandemic, eventually reaching an equilibrium. Greater increases are expected in older individuals which may lead to an increase in healthcare demand due to influenza infections in the coming decades.

#### Introduction

Influenza A H3N2 has circulated in the human population since its emergence in 1968 (the 'Hong Kong' pandemic), causing substantial mortality and morbidity [1]. Infection with a strain of influenza typically causes an immune response, in which antibodies are produced that can recognise and bind to antigenic sites on the virus' surface proteins. These antibodies protect against future infection with antigenically similar strains of influenza. However, due to the accumulation of mutations, new antigenically distinct strains regularly emerge that are able to evade existing immunity ('antigenic drift') [2].

In temperate regions, where influenza transmission is highly seasonal, there are yearly winter epidemics, but during the summer (when there is little transmission) strains regularly become extinct [3]. In contrast, the tropics experience a more continuous transmission regime [4], with more consistent levels of infection throughout the year. As a result, epidemics of new antigenic variants in temperate regions are predominantly seeded by tropical regions [5], particularly East and South-East Asia which has a major role in the production and dissemination of new antigenic variants [6].

Antigenic cartography has been used extensively in assessing the antigenic evolution of influenza H3N2 [1,7]. The antigenic similarity between two strains can be assessed using the haemagglutinin inhibition (HI) assay; this measures the ability of antisera raised against one strain of influenza to prevent agglutination of red blood cells by another strain of influenza (proportional to the antibody concentration against the strain). Through pairwise comparisons of the HI assay across many strains the antigenic distance between strains can be inferred and represented on a two-dimensional 'antigenic map' [1,7].

At the individual level, antibody concentration (against a specific strain) is related to the individual's level of protection against infection (when exposed to that strain) [8] (Fig 1C). An individual's antibody dynamics in response to exposure are complex, depending on their entire infection history. Following infection, antibody responses generated are greatest against the strain of infection, while they are reduced against (antigenically) distant strains. A broadly cross-protective short-term antibody response fades to leave a more specific long-term response [9] (Fig 1A). An individual's first infection with influenza typically elicits the greatest antibody response and subsequent infections elicit progressively reduced antibody responses [10] (Fig 1B). This progressive reduction is known as 'antigenic seniority'.

Epidemic models can be used to explore the dynamics of multi-strain pathogens such as influenza. In these multi-strain models a 'strain space' is defined; this space encompasses all possible strains that can circulate and describes the degree of cross-immunity between any pair of strains [11-13]. Such models have previously been used to explore the evolutionary dynamics of H3N2, incorporating mutation explicitly into the transmission process. However, these models have focused on how mechanisms of immunity can give rise to the evolutionary patterns of H3N2; less consideration has been given to how evolutionary patterns at the global



**Fig 1. Relationship between infection and immunity.** (A) Short- (red) and long-term (blue) components of the antibody response (measured by log-titre of a haemagglutinin inhibition test), and the total response (solid black line), due to an infection with a strain of influenza. Antibody response is greatest against the strain of infection (antigenic distance = 0) and decreases linearly with antigenic distance against other strains. The antigenic distance between two strains quantifies their dissimilarity, with one unit of antigenic distance corresponding to a two-fold dilution in hemagglutinin inhibition (HI) titre. This HI titre measures the ability of antisera raised against one strain of influenza to prevent agglutination of red blood cells by another strain of influenza. The short-term response decreases linearly with time, reaching zero 1.27 years after infection. (B) Long-term components of the antibody response (coloured), and total long-term antibody response (solid black line), due to infection with multiple strains of influenza (evenly spaced in one-dimension of antigenic space). The first infection has the greatest antibody response, decreasing linearly with the number of previous infections. (C) Relationship between the log-titre against a specific strain of influenza, and the protection against infection with that strain. Note that due to our parameterisation (see Methods) the protection against infection does not go to zero for a log-titre of zero. This is not a problem as only the relative values of protection against infection (by log-titre) are of concern — the non-zero value simply decreases the transmission rate which we have nevertheless adjusted to fit observed data.

scale and immune mechanisms at the individual-level influence population-level epidemiological dynamics. Furthermore, the role of antigenic seniority, the magnitude of which has only recently been estimated [9,14], has not yet been investigated.

Here we develop an integrated modelling framework of influenza transmission, immunity, and antigenic evolution to study the long-term epidemiological dynamics of influenza and explore their dependence on different hypothesised and studied mechanisms of immunity [9,14], including antigenic seniority. We simulate the yearly epidemic dynamics of influenza transmission within a temperate region over a 160-year period, following the introduction of a novel influenza virus (i.e. H3N2 dynamics from its emergence in 1968 to 2128). The strains present in the model are defined over a two-dimensional strain space [11,15], analogous to two-dimensional antigenic maps and tuned to accurately resemble the observed H3N2 maps. The degree of cross-immunity between strains is determined by their antigenic distance and a function linking their antigenic distance to immunity, based on previously published work [8,9] (Fig 1). Finally, antigenic evolution is introduced as a global process (no feedback between epidemic dynamics and antigenic evolution), enabling us to explore how evolutionary dynamics, controlled by tropical regions, influence the epidemiological dynamics of influenza in a temperate region unlikely to contribute to global evolutionary dynamics.

#### Results

#### Simulating the long-term dynamics of influenza

We simulated the antigenic evolution of influenza H3N2 using a statistical model (see Methods) that describes the within- and between-year variation in antigenic diversity. The model's parameters were fitted to available data from antigenic maps from 1968 to 2011 (S1 Fig) [1]. Our model allows us to simulate the strains of influenza present in a given epidemic year (N = 20 per year) by their location on a two-dimensional antigenic map (Fig 2). We used the model to simulate many different evolutionary trajectories (S2 Fig) that resemble the punctuated nature of antigenic drift observed in data [7].

Strains in the model — which incorporates seasonality and realistic contact patterns between age-groups (see Methods) — were introduced into a multi-strain agent-based epidemic model at a constant rate. The infection history of each individual in the model was tracked (80,000 individuals) and used to calculate their probability of infection given contact with an individual infected with a specific strain (Figs 1 and S3). Parameter values were selected from previous studies where available or chosen to best match observed data (See Methods and S2 Table). The resulting dynamics resembled the recurrent epidemics of influenza transmission observed in temperate regions (Fig 2). Between years there was substantial variation in the epidemic size, but limited variation in the timing of the epidemic peak (due to seasonality). There was a high degree of correlation between the annual attack rate and the size of the antigenic transitions from the previous year (S4 Fig). Typically, low levels of infection were observed in years with limited antigenic drift and high levels of infection were observed following years with substantial antigenic transitions, though substantial variability year to year means this was not always the case. Following the buildup of population immunity in the years post emergence, epidemics showed little strain diversity during years with substantial antigenic diversity, with the most antigenically distinct strain dominating the epidemic dynamics (see year 4 in Fig 2).

There was substantial variation in the year-to-year annual attack rate (total proportion of population infected) between simulations (S4 and S5 Figs), but good agreement in the overall distribution of yearly attack rates (S5C Fig). For this reason, we considered the dynamics of the average annual attack rate across simulations.

#### Long-term changes in influenza attack rates

The average annual attack rate changed substantially over 160 years since introduction of H3N2 into the population (Fig 3). The highest average attack rate was observed in the pandemic year (mean 47.7% [CrI 47.5%, 48.0%]), the year that the virus was introduced into the human population in which there was (assumed to be) no immunity. Following the pandemic year there was a reduction in attack rate into the second (28.6% [27.8%, 29.4%]) and then third epidemic year (16.9% [15.7%, 18.2%]) in which it reached a minimum. Following this minimum, the average annual attack rate gradually increased, reaching an equilibrium approximately 80 years after the pandemic year (Fig 3A). At 80–89 years since the pandemic





the average annual attack rate was 22.1%, [21.5%, 22.7%], almost 23% greater than the average annual attack rate of 18.0% [17.5%, 18.5%] seen 2–9 years after the pandemic (Fig 3B). This relative increase in the average attack rate over time (and the annual attack rate, including in the pandemic year, and the time taken for the annual attack rate to reach an equilibrium) is sensitive to the assumed average contact rate between individuals in the transmission model with higher (lower) contact rates leading to greater (lesser) relative increases in the attack rate (S7, S8 and S9 Figs). When we assumed a high degree of pre-existing immunity for the population (S7 Fig), while the attack rate in the pandemic year was much lower, the overall long-term dynamics were similar (compared to the model assuming no pre-existing immunity). Additionally, when we assumed that antigenic evolution occurred at a constant rate (i.e. no variation in antigenic transitions between years) the overall long-term dynamics were again similar (S10 Fig).



**Fig 3.** Long-term changes in the average annual attack rate of influenza. (A) Average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0). Estimates are shown for the mean (points) annual attack rate and the 95% confidence interval of the mean (shaded region). (B) The mean annual attack rate (black line) and 95% confidence interval in the mean (shaded region) calculated instead for each decade. The first 'decade' only includes the years 2–9 years after the pandemic (i.e., does not include the first two years in which the attack rate had not yet reached its minimum). (C) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades. Estimates are again shown for the mean (central line) and the 95% confidence interval in the mean (shaded regions). (D) The average annual attack rate by age for the same decades as in (C) but now shown as a proportion of the mean annual attack rate in the pandemic year.

There was significant variation in the average annual attack rates by age (Fig 3). Large differences were observed between age-groups (five-year bands), reflecting differences between their assumed contact rates. In the pandemic year, there was little within-age-group variation in the average attack rate (Fig 3C). This is a result of all individuals initially having the same immune profile (assumed to be no immunity, see S7 Fig for sensitivity analysis). In later years there was within age-group variation. For those aged 0-15 years, there were lower attack rates in older individuals (within an age-group). In those aged over 15 years, there were higher attack rates in older individuals (within an age-group). The pattern in those aged 0-15 years changed little over time (Fig 3D), likely reflecting the relatively quick demographic process causing it; younger individuals have less immunity and so relatively greater attack rates. In contrast, the increase in attack rates for older individuals becomes more pronounced over time, reflecting a slower demographic process; antigenic seniority results in a reduced immune response with each successive infection over time. As time since the pandemic year increases the relative attack rate in older individuals increases to a greater extent, only reaching equilibrium after approximately 80 years. Again, this relative increase in attack rates was sensitive to the assumed average contact rate (S8 and S9 Figs).

#### **Cohort effects**

We identified differences in the number of infections (of influenza H3N2) expected over an individual's lifetime based on their year of birth (Fig 4). Of course, those born before the pandemic year have a smaller number of infections over their lifetime because influenza H3N2 circulated for a lower proportion of their lifetime. Therefore, we only considered cohorts (by year of birth) who experienced an entire lifetime of H3N2 circulation. Individuals born two years after the pandemic had the lowest expected number of lifetime infections, 16.6 [16.4, 16.9], which was only slightly lower than for individuals born in the pandemic year who had 16.9 [16.7, 17.1] infections over their lifetime. Considering only cohorts born three or more years after the pandemic, the expected number of lifetime infections increased slightly with year of birth reaching 17.6 [17.4, 17.7] for those born 80 years after the pandemic, an increase of approximately 6% from the minimum value.

Because of the high infection rates across all age-groups during the pandemic year, an individual's probability of infection during different periods of their life was heavily dependent on their age during the pandemic. Based on broad mixing patterns and the potential consequences of infection, we considered four life stages: early age (0-4 years old), school age (5-17 years old), working age (18-64 years old) and old age (65-79 years old). For all life stages considered, the cohort entering a given life stage in the year of the pandemic had an increased





https://doi.org/10.1371/journal.pcbi.1012893.g004

number of infections expected during that life stage. Similarly, the cohort entering the life stage two years after the pandemic had the lowest number of infections expected during that life stage. These results would be the same for other specified life stages (S11 Fig), reflecting the increased attack rates observed in the pandemic year (and the year after the pandemic), and the relative minimum in attack rates seen two years after the pandemic (i.e., the cohort entering a life stage two years after the pandemic experiences this minimum in attack rates).

Individuals born many years after the pandemic were expected to have a greater number of infections in their old-age. The number of infections an individual was expected to experience during any specific life stage increased with their year of birth. We only considered the cohorts entering a life-stage two years after the pandemic (when the relative minimum in expected number of infections occurred) or later. There were modest increases in the number of expected school age (5-17 years) infections with the cohort born 80 years after the pandemic (last year considered) expecting 3.82 [3.73, 3.91] infections compared to the minimum of 3.56 [3.48, 3.65] infections (cohort born two years before the pandemic). The expected number of working age (18-64 years) and early-age (0-4 years) infections both increased by approximately 15% from their minimum values to 10.9 [10.7, 11.0] and 0.98 [0.94, 1.01] infections respectively for the cohort born 80 years after the pandemic. Finally, the expected number of old-age (65-79 years) infections increased greatly with those born in the year of the pandemic (after which an approximate equilibrium was reached) or later expecting over 30% more oldage infections than the minimum (for those born 63 years before the pandemic). For those born 80 years after the pandemic the expected number of old age infections was 1.91 [1.87, 1.95]. Direct comparisons between the specific life stages we considered are hampered due to their different sized age-ranges, but when considering the cohort dynamics for life stages with equally sized age-ranges (five-year age bands), greater relative increases in the expected number of infections were observed for older-aged infections (S11 Fig).

# The role of antigenic seniority, short-term immunity and long-term immunity

The long-term dynamics of influenza were highly dependent on the assumed degree of antigenic seniority. When no antigenic seniority was assumed, the average annual attack rate reached an equilibrium just a few years after the pandemic, and did not increase thereafter ( $\underline{Fig 5A}$ ). Conversely, a higher degree of antigenic seniority resulted in a greater total increase in the average annual attack rate and a longer time to reach an equilibrium.

Antigenic seniority had little effect on the dynamics over the first ten years, with similar average attack rates for different values of antigenic seniority. However, for higher degrees of antigenic seniority there were greater increases in the average annual attack rate and after 150 years there were substantial differences. With no antigenic seniority the attack rate 150-159 years after the pandemic (S1 Table) was 17.0% [16.5%, 17.4%]. With antigenic seniority representing a 50% reduction in antibody response after 13, 10 and 9 infections (the first integer *Z* such that  $1 - \tau \times (Z - 1) < 0.5$ , where  $\tau$  is the antigenic seniority parameter, and takes values of 0.039 [the empirical estimate], 0.05, and 0.06 respectively), the attack rate 150–159 years after the pandemic was 22.1% [21.6%, 22.7%], 26.9% [26.1%, 27.6%] and 39.5% [38.3%, 40.7%] respectively. The increase in attack rate was again proportionally higher in older individuals (Figs 5C and S12).

The average annual attack rate was highly dependent on short- and long-term antibody responses. Increasing the strength of long-term immunity, or the duration of short-term immunity, led to lower levels of the average annual attack rate (Figs 6 and S13). This in turn had a feedback effect on the long-term dynamics; when the average attack rate is higher individuals are infected more often over their lifetime and so antigenic seniority has a more pronounced effect, driving greater increases in the attack rate over time.





Comparing across simulations we were able to identify an autocorrelation structure (<u>S6</u> Fig). We measured the correlation of annual attack rates between epidemic years. For the empirical estimate of all immunity parameters, there was a significant level of negative correlation, -0.15 [-0.16, -0.14], between adjacent years; large epidemics were more likely to be followed by small epidemics. In contrast, epidemics separated by two years had a significant level of positive correlation, 0.10 [0.09, 0.11]; large epidemics were more likely to be followed by large epidemics two years later (probably due to small epidemics being more likely in the year between). There was little difference between the correlations for epidemics separated by 3–20 years, with a small level of positive correlation of approximately 0.01. This autocorrelation structure was approximately stable from 20 years after the pandemic (S6 Fig).

Changing the duration of short-term immunity influenced the auto-correlation structure between epidemics (Fig 6B). There was a greater level of negative correlation between adjacent epidemics when the duration of short-term immunity was longer. The relationship was more complex for epidemics separated by multiple years; for example, for epidemics separated by four years, increasing the duration of short-term immunity to two years (the empirical estimate is 1.27 years) led to positive correlation but increasing the duration of short-term immunity to 3 years led to negative correlation. The changes in autocorrelation structures were only apparent in the average annual attack rates during the first ten years of the simulation (Fig 6A).



**Fig 6.** The role of short-term immunity in the epidemiological dynamics of influenza A H3N2. (A) The average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0), for different durations of short-term immunity (time before short-term immunity goes to zero). Estimates are shown with mean (points) and central 95% confidence intervals (shaded region). (B) The average Pearson correlation in the annual attack rate between epidemic years (as a function of the number of years between epidemics), estimated over the epidemic years 20–159 years after the pandemic. Estimates are shown with their mean (points) and 95% confidence interval (bars) for different durations of short-term immunity (colours, see legend). The dashed black line high-lights a correlation of 0 (no correlation)..

#### Discussion

By integrating models of transmission, antigenic drift, and immunity, we have simulated the long-term dynamics of influenza H3N2 in temperate regions. While previous studies have focused on the evolutionary dynamics of influenza [11–13], we have explored the long-term epidemiological dynamics of H3N2 transmission. Our analysis considers influenza dynamics over a 160-year timescale, reflecting circulation of H3N2 since its emergence in 1968 during the Hong Kong pandemic through to the year 2128. Our model captures the stochastic nature of influenza transmission with high annual variability in the epidemic attack rate. We observed that on average the annual attack rate should have been increasing since 1970 (two years after the pandemic), with relatively greater increases in the annual attack rate for older individuals. This is difficult to verify empirically, as data on influenza infections have not been collected consistently since H3N2's emergence. And what data are available, even over recent decades, are often heavily biased towards individuals seeking healthcare (who are not representative of the overall population of infected individuals) [16] and exhibiting influenza-like illness, a symptom definition that is non-specific to influenza (the majority of influenza-like illness is not attributable to influenza infection) [17,18].

Our findings have implications for public health. If the average annual attack rate for H3N2 has still not reached an equilibrium, as suggested by our analysis (where the equilibrium is expected only after 80 years, i.e., around the year 2048), then increases in the average annual attack rate may be expected in the coming decades. This could drive an increase in the future healthcare demand due to H3N2 infections, which would be compounded by relatively greater increases in infections for older individuals, for whom the risk of severe illness is highest [19]. Additionally, our analysis highlights differences between individual cohorts: someone

born in 1968 (or later) was expected to have over 30% more infections in their old age (65–79 years) than someone who was 63 in the year of the pandemic (born in 1905). Future individual risk to severe (old age) infections has thus already reached its maximum in those currently aged under 55 years old.

The model makes several simplifying assumptions. Countries' age-distributions vary due to demographic factors [20]. Our results, which assume a uniform age-distribution, must be considered in this context given the sharp increase in severity of infection with age [21]. Agedistributions may also change over time (with many countries having ageing populations), influencing epidemic dynamics [22]. This could potentially introduce more complex cohort effects in the context of the multi-strain transmission dynamics considered here. Similarly, all contact rates were modelled as constant over time; a long-term trend of increasing contact rates would lead to additional upwards pressure on annual influenza attack rates (and vice versa). There have been some short-term reductions in contact rates due to interventions introduced during the SARS-CoV-2 pandemic [23], these measures reduced influenza transmission for much of 2020–2022 [24]. The potential cohort effects for H3N2 epidemiology introduced by responses to COVID-19 will likely not be immediately apparent. Our model considers only infection with influenza H3N2, but there have been many respiratory viruses (e.g, influenza H1N1) co-circulating which can interfere with transmission [25,26]; this might have acted to reduce H3N2 attack rates from what would have been observed without their circulation/interference and may have introduced additional cohort effects, particularly when pathogens emerged after H3N2 began circulating (e.g., influenza H1N1).

We found that the long-term dynamics of influenza H3N2 were highly dependent on the degree of antigenic seniority. Higher levels of antigenic seniority resulted in a greater increase in the average annual attack rate. If there was no antigenic seniority, then the average annual attack rate would have reached an equilibrium shortly after the pandemic and would not have increased thereafter. We used a simple parameterisation of antigenic seniority using the results of Kucharski et al [9]. This is the only available parameterisation of antibody dynamics that includes antigenic seniority and relates the antibody response between two strains to their antigenic seniority used an exponential function [14] but given the expected total number of infections of an individual in our model (approximately 17–18 over a lifetime), the two functions would not be expected to produce substantially different results (these two choices of functional forms are more divergent at greater numbers of infections). Both parameterisations are based on serology studies collecting data approximately 40 years after the pandemic, and so the validity of these models after 80 years' of virus circulation is unclear; though this would be a problem for any recent dataset given the long timescale of our analysis.

The magnitude of effect that antigenic seniority had on the long-term dynamics of influenza H3N2 was heavily modulated by the average annual attack rate in our model. Changing the overall level of immunity conferred by infection — either by increasing the strength of long-term immunity or by reducing the waning rate of short-term immunity — led to a change in the baseline average attack rate. With higher annual attack rates, individuals had a greater number of infections over their lifetime, and so antigenic seniority had a greater impact on the long-term dynamics. The average annual attack rate of influenza H3N2 is poorly characterised. We selected model parameters so that the average annual attack rate over the first 40 years of simulations was approximately 20%. This is consistent with results from analysis performed using longitudinal serology data on a cohort in Vietnam [9], although estimates of the attack rate from another Vietnamese cohort were higher at 25.6% [27]. However, it is not clear if this is an appropriate value for the temperate regions considered in this paper. Some estimates available for temperate regions have been lower [28,29], but the methodology used primarily identified symptomatic cases and it is known that a high proportion of influenza infections are asymptomatic [30]. Studies using the 'gold standard' method for identifying recent influenza infection (serologic detection of a four-fold increase in antibody titres) have estimated greater overall annual attack rates for seasonal influenza (H3N2, H1N1, influenza B) of 19% in the United Kingdom [30] and 24% in New Zealand [31], of which H3N2 should be a significant proportion [5]. However, another study has suggested that even when these 'gold-standard' methods are used, they could be underestimating attack rates by as much as 27% [32].

Our model assumed that antigenic drift occurs at a constant rate. If the rate of antigenic drift increased (or decreased) we would expect an increase (decrease) in the average attack rate due to the more frequent (less frequent) emergence of antigenically diverse immune- evading strains. Though a single temperate region likely has little impact on the global dynamics of antigenic drift, there may be a feedback effect between global population immunity and long-term antigenic drift which our model does not account for. For example, long-term increases in the attack rate due to antigenic seniority could lead to an increased rate of antigenic drift (more transmission so more opportunities for new strains to emerge) [33]. Alternatively, the dampening of immune responses due to antigenic seniority could reduce (within-host) immune pressure and decrease the rate of antigenic evolution [34]. Further work is required to quantify if and how the global rate of antigenic drift is likely to change over the long-term. Additionally, while our model allows significant strain diversity within any given season, it does not allow strain diversity to increase over time. Therefore, our model could not capture long-term divergences in antigenic evolution (for example, the emergence of bimodal distributions of antigenic distances), as has been observed for influenza B which diverged into two distinct lineages in the 1980s both of which continued to circulate (as two distinct evolutionary pathways) for decades [35]. This may be relevant in the future for H3N2 (e.g. see emergence and circulation of antigenically diverse clades of H3N2 from 2012-2022 [36])".

We did not consider the effect of vaccination. Influenza vaccines are updated regularly to better match circulating strains [37], and are deployed annually in many countries with varying rates of coverage [38]. While vaccination can prevent illness, hospitalisations, and deaths (when the vaccine strain is a good match to circulating strains) [39,40], the effect on short-term transmission dynamics in the population may be minimal: subsets of the population with higher vaccination coverage (older individuals) [41] often contribute less to transmission [42]; vaccines have limited effectiveness when they match poorly to the circulating strains [39], which is a regular occurrence for H3N2 due to its rapid antigenic evolution; and though vaccines can be effective at preventing symptomatic infection, their effectiveness against asymptomatic infection is poorly understood [43]. However, vaccination on an individual's long-term immune response is not well understood. This includes the degree to which vaccination influences antigenic seniority. We note that some work has identified reduced antibody responses following vaccination, in individuals who are vaccinated frequently [44].

Although our model has been calibrated for influenza H3N2, it could be generalised to other pathogens that have similar dynamics in which antigenic drift sustains long term transmission. Influenza H1N1pndm09 has circulated in the human population since the 2009 'Swine flu' pandemic and likely exhibits similar mechanisms of immunity [45] (though this has not been confirmed). During the 2009 pandemic there were more complex patterns of pre-existing immunity to H1N1pdm09 due to: 1) the circulation of other H1N1 strains from 1977–2009 (going extinct following the 2009 pandemic) [1,46]; and 2) the antigenic similarity of the H1N1pdm09 strain to the 1918 pandemic H1N1 strain [2], meaning that older

individuals (who may have been infected with more antigenically similar strains) had higher levels of pre-existing immunity [47]. It is thus unclear at what stage of its long-term dynamics the H1N1pdm09 virus is currently at. Is it already at an equilibrium state due to high levels of previous exposure to H1N1 viruses? Or given it is less than 15 years since the pandemic, is the annual attack rate still at a relative minimum? The dynamics of SARS-CoV-2 are even less well understood. Since its introduction to the human population in late 2019, we have seen the frequent emergence of new immune-evading variants. A preliminary analysis using antigenic cartography has identified that these new variants are antigenically distinct to previous variants [48]. However, it is still unclear what long-term dynamics of immunity and evolution will be exhibited by SARS-CoV-2 and if they will resemble that of influenza H3N2. Recent serological studies have demonstrated dampened immune responses due to repeat exposure to SARS-CoV-2 [49,50], hinting at the presence of antigenic seniority, but this requires investigation over a longer time frame.

Our study highlights the need for data. Following a pandemic, our analysis suggests that a relative minimum in the annual attack rate may be observed, but if there is any degree of antigenic seniority governing the immune response to future infections, then the average annual attack rate will increase over demographic timescales. Studies are needed to assess the dynamics of H1N1pdm09 and SARS-CoV-2 infections and immunity so that any long-term rise in infection rates can be anticipated. Quantifying the interactions between vaccination and long-term immunity is crucial (including for H3N2) for identifying cohorts that may be at greater risk of infection in the future. A better understanding of the effect of vaccination on long-term immunity would also allow vaccination strategies to be optimised against short-and long-term public health objectives at both the population- and individual-level. Developing long-term surveillance strategies for respiratory viruses [51], that consistently estimate population infection levels, antigenic evolution and immunity, would allow the long-term dynamics to be better quantified and public health strategies to be refined.

#### Methods

#### Global drift model

We use a statistical model to simulate the antigenic drift of influenza. The model aims to replicate (phenomenologically) the observed punctuated nature of influenza H3N2's antigenic evolution, but it does not seek to explain the mechanisms that lead to such observations and so can not predict changes in evolutionary patterns into the future. The antigenic drift of influenza was simulated across a two-dimensional strain space, analogous to an antigenic map. Individual strains were defined by their two-dimensional antigenic coordinates,  $\underline{s_i} = (s_{i,x}, s_{i,y})$ . In a given year, N, strains were assumed to be normally distributed around a mean coordinate,  $\psi_n = (\psi_{n,x}, \psi_{n,y})$ , with standard deviation,  $\rho$ :

$$S_{\underline{n},\underline{i}} \sim N(\underline{\psi}_{\underline{n}}, \rho).$$

Between years the mean coordinate drifts stochastically:

$$\underline{\psi_n} = \underline{\psi_{n-1}} + \underline{u_n}.$$

Where,  $\underline{u_n}$ , is a random variable. The x-component of  $\underline{u_n}$  is drawn from an exponential distribution,  $u_{n,x} \sim Exp(\zeta)$ , and the y-component of  $\underline{u_n}$  is drawn from a normal distribution,  $u_{n,y} \sim N(0,\eta)$ . This formulation ensures that the principal component of antigenic drift is along the x-axis, while still allowing variation across both dimensions. This model of antigenic drift can be defined by the three parameters:  $\rho$ ,  $\zeta$ , and  $\eta$ .

In order to perform stochastic simulations of H3N2's antigenic drift we first estimated the parameter values. We implemented the above model in STAN and sampled from the parameter posterior using a No-U-Turns Sampler [52]. In the implementation of the model the random variable distributions were the likelihood. All parameters were given an uninformative constant prior. The model was fitted to an antigenic map for influenza H3N2 covering the years 1968-2011 [1] (S1 Fig). Parameter estimates were taken as the mean value of parameter posteriors:  $\rho = 1.14$ ;  $\zeta = 0.99$ ;  $\eta = 1.36$ .

We also performed a sensitivity analysis assuming that the mean antigenic coordinates drifted along the x-axis at a constant rate (by  $1/\zeta$  antigenic units each year) and that there was no drift along the y component (i.e.  $\eta = 0.0$ ).

#### Epidemiological model

The long-term epidemiological dynamics of influenza were simulated using an agent-based multi-strain transmission model following a formulation of the SIS model [53]. At any point in time, an individual (agent) was either susceptible or infected (and infectious) (with a specific strain). Discrete epidemic years (each one 365 days long) were simulated with daily time-steps. During each timestep, either a susceptible individual could become infected following contact with an infected individual, or an infected individual could clear their infection and re-enter the susceptible class (though the individual would now have immunity to infection with antigenically similar strains, see below). Infection histories (strain and timing of infections) were tracked for all individuals.

Infected individuals cleared their infection (reentered the susceptible class) at a constant rate  $\gamma$ . During each timestep the number of infected individuals who cleared their infection was given by a random number chosen from a Poisson distribution,  $Pois(I_t \times \gamma)$ , where  $I_t$  was the number of infected individuals on the given day. Infected individuals were selected to clear their infection at random. The parameter  $\gamma$  was chosen to have a value of 0.33 to give an average duration of infection of 3 days.

Contact between individuals was mediated by their age-groups. At the beginning of the simulation each individual was assigned an integer age with equal proportions aged 0 to 79 inclusive (1000 individuals each age for a total population of 80,000). Age-groups were defined as 5-year age bands (0–4 years, 5–9 years, etc.). Relative contact rates between age-groups were calculated using the R package commat [54,55]. The relative contact rates were scaled so that the maximum contact rate was equal to 1 (the mean contact rate was 0.0742) (S2 Table). Contacts between infectious individuals and susceptible individuals were then calculated using a similar approach to Tsai et al [56]. The maximum number of contacts was given

by a random number chosen from a Poisson distribution,  $Pois\left(I_t \times \frac{S_t}{N} \times \beta_t\right)$ . Where  $\frac{S_t}{N}$  is the number of infectious individuals at time t,  $\frac{S_t}{N}$ . is the proportion of the population susceptible and  $\beta_t$  is the seasonally forced maximum contact rate.  $\beta_t$  is assumed to be sinusoidally varying with a period of one year, an average value of  $\beta_0$  and a maximum value of  $\beta_0 + \beta_1$ :  $\beta_t = \beta_0 \times (1 - \beta_1 cos(2\pi t / 365))$ . Given a maximum number of contacts, pairs of infectious and susceptible individuals are randomly selected. The probability of an infectioevent in a pair is then given by:

$$P(infection)_{ij} = M_{ij} \times (1 - P_{ij})$$

Here the probability that susceptible individual *j* is infected by infectious individual *i* is controlled by  $M_{ij}$ , the contact rate between *i* and *j*'s age-groups, and  $P_{ij}$ , the susceptibility of individual *j* to the strain infecting individual *i* (see Immunity model below).

Strains are introduced into the epidemic model from the global pool model. Each daily timestep a random number of susceptible individuals are selected using a Poisson distribution,  $Pois(S_t \times \chi)$ , where  $\chi$  is the importation rate parameter and was chosen to be small ( $\chi = 0.00005$ ). Each individual selected is then infected with a strain (randomly selected from the global pool) with probability  $1 - P_{ij}$ . Where  $P_{ij}$  is again individual *j*'s protection against infection with strain *i*.

Between epidemic years: (1) the global drift model is updated (see above), (2) each individual's age increases by 1 and those that should have turned 80 are reset to 0 year olds with no infection history (birth and death), and (3) all current infections are cleared. Clearing infections is done so that strains from previous years are removed when the global drift model is updated. At this point in the epidemic year (end/beginning) seasonality was at its lowest and so infection levels are low.

#### Immunity model

An individual's protection against infection with a specific strain is determined by their entire infection history. An individual's antibody concentration (measured by log-titre) against a specific strain is determined using a model presented in Kucharski et al [9]. Each infection in an individual's infection history produces an antibody response. The log-titre of the antibody response is controlled by multiple components:

- 1. Long-term boosting. Infection with a strain *i* will produce a fixed log-titre against that strain, with parameter value  $\mu_1$ .
- 2. Short-term boosting. Infection with a strain *m* will produce an additional contribution to the log-titre against the strain that wanes over time. The contribution is given by  $\mu_2 w(m) = \mu_2 \times max\{0, 1 \omega t_m\}$ , where  $\mu_2$  is the boosting parameter,  $\omega$  is the waning parameter, and  $t_m$  is how much time has elapsed since infection with strain *m*.
- 3. Cross-reactivity. Both long- and short-term boosting will also produce log-titres against antigenically similar strains. The log-titre against an infecting strain *j*, due to an infection with strain *m* is given by  $d(m, j) = max\{0, 1 \sigma \times \delta_{mj}\}$ , where  $\delta_{mj}$  is the euclidean distan between strains *m* and *j* in two-dimensional antigenic space, and  $\sigma$  is a parameter that has different values for the cross-reactivity due to long-term boosting ( $\sigma_1$ ) and due to short-term boosting ( $\sigma_2$ ).
- 4. Antigenic seniority. An individual's initial infection produces the response described above. For subsequent infections the antibody response is progressively suppressed. The log-titre produced by infection with strain *m* is scaled by the factor  $s(X,m) = max\{0, 1 \tau(N_m 1)\}$ , where  $N_m$  is the number of strain *m* in the infection history (e.g., the first strain is 1, the second is 2),  $\tau$  is a further parameter of the model and *X* is the set of all strains in an individual's infection history.

The overall log-titre against strain *j*, due to an individual's entire infection history is then given by:

$$\lambda_{j} = \sum_{m \in X} s(X,m) \times \left[ \mu_{1}d_{1}(m,j) + \mu_{2}w(m)d_{2}(m,j) \right]$$

This is controlled by the parameters:  $\mu_1$ ,  $\sigma_1$ ,  $\mu_2$ ,  $\sigma_2$ ,  $\omega$ , and  $\tau$ . We take the values of these parameters from Kucharski et al. [9]:  $\mu_1 = 2.02$ ;  $\sigma_1 = 0.130$ ;  $\mu_2 = 2.69$ ;  $\sigma_2 = 0.031$ ;  $\omega = 0.79$ ;  $\tau = 0.039$ .

We convert the log-titre to protection against infection using the model presented in Coudeville et al. [8]. The protection against infection is given by:

$$Protection(T_{j}) = 1 - \frac{1}{1 + exp\left\{ \in \left( log(T_{j}) - \alpha \right) \right\}}$$

Here  $\epsilon$  and  $\alpha$  are parameters, and  $T_j$  is the titre against strain *j*. The conversion between  $T_j$  and  $\lambda_j$  is slightly non-trivial due to the different definitions used in the two studies considered and is given by:  $T_j = 2^{\lambda_j - 1} \times 10$ . The parameter values are taken from Coudeville et al. [8]:  $\epsilon = 1.299$ ;  $\alpha = 2.844$ .

#### **Parameter selection**

Before the long-term epidemiological dynamics of influenza transmission could be simulated, parameter values for the epidemiological model had to be selected. The parameter values have all been stated in the previous sections (and S3 Table) with the exception of the average contact rate,  $\beta_0$ , and the degree of seasonal forcing,  $\beta_1$  (maximum contact rate of  $\beta_0 + \beta_1$ ) ). We investigated the transmission dynamics for different values of these parameters. We performed 20 simulations of the first 40 years of influenza epidemics for each parameter set. For each parameter set, we measured the average annual attack rate over all simulations and years, and the variation in timing of the week in which influenza incidence was highest (peak week). We then used a generalised additive model implemented in the R package mgcv [57] to predict the value of these quantities across the continuous parameter space (S14 Fig). We selected parameter values so that the average annual attack rate was approximately 20% [9] and that the standard deviation in the peak week in years 21 to 40 was approximately 3.5 weeks (matching the standard deviation in peak week timing from 1991 to 2003 averaged across 48 US states) [58]. The selected parameter values for simulating the long-term dynamics of influenza were  $\beta_0 = 5.5$  and  $\beta_1 = 0.25$ . As a sensitivity analysis we also ran simulations with  $\beta_0 = 5$  and  $\beta_0 = 5.75$ ; these parameters were chosen to give an average annual attack rate (in the first 40 years) of approximately 15% and 22.5% respectively.

#### Statistical analyses

We simulated 160 years of influenza transmission. The main analyses were performed with the parameters given in <u>S2 Table</u>. Sensitivity analyses were also performed assuming different values for the antigenic seniority parameter ( $\tau$ ), the duration of short-term immunity ( $1/\omega$ ), the strength of long-term immunity ( $\mu_1$ ), the average contact rate ( $\beta_0$ ) and for a uniform contact matrix (all contact rates between age-groups the same, equal to the mean contact rate of the contact matrix used in the main analysis) (<u>S3 Table</u>). For each set of parameters, we ran 256 simulations (each lasting 160 years).

Additionally, a sensitivity analysis was performed to explore the effect of pre-existing population immunity at the time of H3N2 emergence due to circulation of other influenza viruses in the population (e.g. H2N2). To simulate pre-existing immunity, we assumed that at the beginning of the simulation all individuals exhibited the long-term antibody response associated with an infection at the antigenic coordinates (0,0), the mean antigenic coordinates in the pandemic year. This gives the population a high degree of pre-existing immunity in the pandemic year.

We also performed a sensitivity analysis assuming that mean antigenic coordinates drifted along the x-axis by a constant amount each year (i.e. constant evolutionary rate of antigenic evolution).

For each epidemic year (in each simulation) we measured the annual attack rate as the total number of infections in the year divided by the total size of the population (80,000). Similarly, we measured the annual attack rate for each integer age as the number of infections in

individuals of that age divided by the number of individuals of that age (1000). We calculated the mean annual attack rate (for each year, decade and by age for specific years and decades) across simulations. The standard deviation in this mean was estimated by measuring the standard deviation in the distribution (across simulations and years [for decades]) and dividing by the square root of the sample size. The 95% confidence interval for the mean annual attack rate was estimated by subtracting/adding 1.96 times the standard deviation in the mean.

For each epidemic year (after the pandemic year) we measured the Pearson correlation (and 95% confidence interval) between the annual attack rate (in the given year) and the size of the antigenic jump from the previous year across the 256 simulations. The size of the antigenic jump between years was measured as the distance (in antigenic space) between the mean antigenic coordinates of adjacent years. We also considered the Spearman correlation coefficient and obtained similar estimates (S4 Fig)

For each epidemic year we measured the Pearson correlation between the annual attack rate (in the given year) and the annual attack rate 1 to 20 years later across the 256 simulations. The standard deviation is calculated for the Fisher's transformation of the correlation coefficient. The 95% confidence interval is then estimated by subtracting/adding 1.96 times the standard deviation and then performing the inverse Fisher's transformation. In the first twenty years of epidemics the correlation between years changed slightly over time before reaching an equilibrium. We calculated the average correlation over all epidemic years 20 years after the pandemic year; all correlation coefficients underwent Fisher's transformation, the mean and the standard deviation in the mean was calculated, the mean and 95% confidence intervals (+/-1.96 times the standard deviation) then underwent the inverse Fisher's transformation. We also considered the Spearman correlation coefficient and obtained similar estimates (compare Figs 6B and S15).

#### Supporting information

**S1 Fig. Fit of the global drift model to data from an antigenic map. Antigenic map for influenza H3N2 covering the years 1968–2011, taken from Bedford et al 2014** [1]. The antigenic distance between two strains (points) quantifies their dissimilarity, with one unit of antigenic distance corresponding to a two-fold dilution in hemagglutinin inhibition titre. Strains (points) are coloured by the year in which they were sequenced, and some years have been explicitly labelled for clarity. The statistical model describing global antigenic drift (see Methods) has been fitted to this antigenic map. The posterior of the mean antigenic coordinates, n = (n,x, n,y), for each year are shown with the posterior mean (centre of cross) and 95% confidence interval in antigenic dimension 1 (horizontal line of cross) and antigenic dimension 2 (vertical line of cross). Grey lines connect the mean antigenic coordinates between adjacent year; note that the mean antigenic coordinates increase along antigenic dimension 1 each year (from left to right). The strains associated with a given year are connected to the mean antigenic coordinates for that year by dashed lines, and are coloured the same. Note that in some years there are no strains on the antigenic map (e.g., 1978 [the second light blue year]). (TIFF)

**S2 Fig. Example simulations of the global drift model. Four independent simulations of the global drift model.** The simulated antigenic diversity of influenza strains for 44 years (colours highlight separate years with 10 strains per year). Strains are located on a two-dimensional antigenic map. The antigenic distance between two strains quantifies their dissimilarity, with one unit of antigenic distance corresponding to a two-fold dilution in hemagglutinin inhibition titre. Strains in a single year (points) are normally distributed around mean antigenic coordinates (dotted lines connect strains in a given year to the mean antigenic

coordinates). The mean antigenic coordinates change each year (grey line connecting adjacent years), always progressing along antigenic dimension 1 (left to right). The mean antigenic coordinates in the first year were set to be (0, 0). (TIFF)

S3 Fig. Example individual-level immunity due to multiple infections with antigenically distinct strains. (Top panel) One example simulation (see S2 Fig) of the antigenic diversity of influenza strains (points) for 44 years (colours highlight separate years with 10 strains per year). Strains are located on a two-dimensional antigenic map. The antigenic distance between two strains quantifies their dissimilarity, with one unit of antigenic distance corresponding to a two-fold dilution in hemagglutinin inhibition titre. The simulated infections of a single individual are indicated by crosses. The infection furthest along antigenic dimension 1 is the most recent infection. (Middle and bottom panels) The estimated protection against infection from a strain of influenza, as a function of the strain's location on the antigenic map. The middle panel shows the protection against infection immediately following the most recent infection (red cross), a combination of the long-term immune responses of all past infections (black crosses and red cross), and the short-term immune response of the most recent infection. The bottom panel shows the protection against infection a long time after the most recent infection (any time greater than 1.27 years: the time taken for the short-term immune response to go to 0), a combination of only the long-term immune responses of all past infections.



S4 Fig. Correlation between annual attack rates and the size of antigenic transition. (Top panel) Point estimates (dots) and 95% confidence interval (error bars) of the Pearson correlation (across 256 simulations) between the annual attack rate and the size of the antigenic jump from the previous epidemic year. The size of the antigenic jump between years was measured as the distance (in antigenic space) between the mean antigenic coordinates of the adjacent years. (Middle panel) Same as above but for the Spearman correlation. (Bottom panels) The annual attack rate plotted against the size of the antigenic jump for all 256 simulations shown for four different epidemic years. For visualisation of the relationship between the two variables we have also plotted a linear regression line (fitted using ordinary least squares).



**S5 Fig. Variation in the annual attack rate between simulations.** (A) The annual attack rate for each epidemic year (points), including the pandemic year (years since pandemic = 0), shown for four independent simulations (four unique colours and lines). (B) The average (mean) annual attack rate for each epidemic year (points) and the central 50% (dark blue), 80% (blue), and 95% (light blue) tolerance intervals, estimated from 256 simulations. Note that the black line and points are the same as in Fig 3. (C) Probability density of the annual attack rate over all 160 epidemic years of each simulation. The probability density was calculated for all 256 simulations; the median (black line) and 50%, 80%, and 95% pointwise confidence intervals (shaded regions) are shown. (D) Probability distributions (grey shaded regions) of the annual attack rates (points) and their 95% confidence interval (small black bars) for each decade. (TIFF)

**S6 Fig.** The correlation structure of the attack rate between epidemic years. (A) The correlation (points) between the attack rate in a single epidemic year (x-axis) and the attack rate

in a future epidemic year (subheadings give time between epidemic years). Each estimate of correlation (points), and its 95% confidence interval (grey lines) is made across 256 simulations. (B) The average correlation of the attack rates between epidemic years (as a function of the number of years between epidemics), estimated over the epidemic years 20–159 years after the pandemic. Estimates are shown with their mean (points) and 95% confidence interval (black bars). The dashed red line in both subplots highlights a correlation of 0 (no correlation).

(TIFF)

**S7 Fig. Long-term changes in the average annual attack rate of influenza. Sensitivity analysis with pre-existing population immunity.** The same as in Fig.3, but simulations were instead run with all individuals having a long-term immune response against the mean antigenic coordinates for the pandemic year (see Methods). (A) Average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0). Estimates are shown for the mean (points) annual attack rate and the 95% confidence interval in the mean (shaded region). (B) The mean annual attack rate (black line) and 95% confidence interval in the mean (shaded region) calculated instead for each decade. The first 'decade' only includes the years 2–9 years after the pandemic (i.e., does not include the first two years in which the attack rate was still at higher levels). (C) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades. Estimates are again shown for the mean (central line) and the 95% confidence interval in the mean (shaded regions). (D) The average annual attack rate by age for the same decades as in (C) but now shown as a proportion of the mean annual attack rate in the pandemic year. (TIFF)

**S8 Fig. Long-term changes in the average annual attack rate of influenza. Sensitivity analysis with higher transmission rate.** The same as in Fig\_3, but simulations were instead run with parameters chosen so the average annual attack rate in the first 40 years was 22.5% (see Methods). **(A)** Average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0). Estimates are shown for the mean (points) annual attack rate and the 95% confidence interval in the mean (shaded region). **(B)** The mean annual attack rate (black line) and 95% confidence interval in the mean (shaded region) calculated instead for each decade. The first 'decade' only includes the years 2–9 years after the pandemic (i.e., does not include the first two years in which the attack rate was still at higher levels). **(C)** The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades. Estimates are again shown for the mean (central line) and the 95% confidence interval in the mean (shaded regions). **(D)** The average annual attack rate by age for the same decades as in (C) but now shown as a proportion of the mean annual attack rate in the pandemic year. (TIFF)

**S9 Fig. Long-term changes in the average annual attack rate of influenza. Sensitivity analysis with lower transmission rate.** The same as in Fig.3, but simulations were instead run with parameters chosen so the average annual attack rate in the first 40 years was 15.0% (see Methods). (A) Average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0). Estimates are shown for the mean (points) annual attack rate and the 95% confidence interval in the mean (shaded region). (B) The mean annual attack rate (black line) and 95% confidence interval in the mean (shaded region) calculated instead for each decade. The first 'decade' only includes the years 2–9 years after the pandemic (i.e., does not include the first two years in which the attack rate was still at higher levels). (C) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades.

Estimates are again shown for the mean (central line) and the 95% confidence interval in the mean (shaded regions). **(D)** The average annual attack rate by age for the same decades as in (C) but now shown as a proportion of the mean annual attack rate in the pandemic year. (TIFF)

S10 Fig. Long-term changes in the average annual attack rate of influenza. Sensitivity analysis with constant rate of antigenic evolution. The same as in Fig 3, but simulations were instead run with a global drift model that assumed mean antigenic coordinates drifted along the x-axis at a constant rate with no drift along the y-axis (i.e. no variation in antigenic transitions between years). (A) Average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0). Estimates are shown for the mean (points) annual attack rate and the 95% confidence interval in the mean (shaded region). (B) The mean annual attack rate (black line) and 95% confidence interval in the mean (shaded region) calculated instead for each decade. The first 'decade' only includes the years 2–9 years after the pandemic (i.e., does not include the first two years in which the attack rate was still at higher levels). (C) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades. Estimates are again shown for the mean (central line) and the 95% confidence interval in the mean (shaded regions). (D) The average annual attack rate by age for the same decades as in (C) but now shown as a proportion of the mean annual attack rate in the pandemic year. (TIFF)

**S11 Fig. The expected number of infections by year of birth. (Left) The expected number of infections an individual will experience during different life-stages (5 year age ranges) by their year of birth (cohort).** Estimates are only shown for the cohorts in which influenza was circulating all years in which they were in the life-stage. For example, for the life-stage representing infections while aged 20–24 years, only those born 20 years before the pandemic and later are considered. Horizontal dashed lines show the maximum expected number of infections, found for the cohort entering the life-stage in the year of the pandemic (red), and the minimum expected number of infections, found for the cohort entering the life-stage two years after the pandemic (black). (Right) The expected number of infections an individual will experience during different life-stages by cohort relative to the cohorts for which the value (for each life-stage) is a minimum (black dashed line). All estimates are shown with the mean (lines) and the 95% confidence interval in the mean (shaded region). (TIFF)

S12 Fig. The effect of antigenic seniority on the long-term dynamics of influenza A H3N2. Sensitivity analysis using uniform contact rates. (A) and (B) are the same as in Fig 4, but the age-contact matrix now assumes equal contact rates between all age-groups (the mean contact rate matches the mean contact rate of the age-contact matrix used in the main analysis). (A) The average annual attack rate for each epidemic year, including the pandemic year (years since pandemic = 0), for different values of the antigenic seniority parameter (see Methods). The number of infections until the antibody response is suppressed by 50% is given by 0.5 divided by the antigenic seniority parameter. Estimates are shown with the mean (points) and the 95% confidence interval for the mean (shaded region). (B) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades, for different values of the antigenic seniority parameter. Estimates are shown with the mean (central line) and the 95% confidence interval for the mean (shaded regions). (TIFF)

S13 Fig. The effect of long-term immunity on the long-term dynamics of influenza A H3N2. (A) The average annual attack rate for each epidemic year, including the pandemic

year (years since pandemic = 0), for different values of the parameter setting the maximum long-term antibody response, 1(see Methods). Estimates are shown with the mean (points) and the 95% confidence interval for the mean (shaded region). (B) The average annual attack rate by age (0–79 years) for the pandemic year, and for specific decades, for different values of the long-term immunity parameter. Estimates are shown with the mean (central line) and the 95% confidence interval for the mean (shaded regions). (C) The average annual attack rate by age for the years 150–159 years after the pandemic, shown relative to the average annual attack rate by age in the year of the pandemic, for different values of the long-term immunity parameter. Estimates are again shown with the mean (central line) and the 95% confidence interval for the pandemic, for different values of the long-term immunity parameter. Estimates are again shown with the mean (central line) and the 95% confidence interval for the pandemic, for different values of the long-term immunity parameter. Estimates are again shown with the mean (central line) and the 95% confidence interval for the mean (shaded regions). (TIFF)

**S14 Fig. Selection of parameter values for the contact rate and seasonality.** (A) The average annual attack rate over the first 40 epidemic years (including the pandemic year). (B) The standard deviation in the timing of the peak week over the first 40 epidemic years (including the pandemic year). (A,B) Point estimates were made using the epidemiological model (20 simulations) at set values (grid of black dots) of the contact rate parameter, 0, and the seasonality parameter, 1. Continuous estimates were then made using generalised additive models, with the output variable the point estimates, and the parameter values (in a two-dimensional thin plate spline) as the input variables. Contours are shown (white) for specific values of the contant in the first 40 years the average annual attack rate was approximately 20% and the standard deviation in the timing of the peak week was approximately 3.5 weeks. Parameters for sensitivity analysis (orange triangles) were chosen so that in the first 40 years the average annual attack rate was approximately 3.5 weeks. Parameters for sensitivity analysis (orange triangles) were chosen so that in the first 40 years the average annual attack rate was approximately 15% and 22.5%. (TIFF)

**S15 Fig. The role of short-term immunity in the correlation between annual attack rates of influenza H3N2. Sensitivity analysis using the Spearman correlation coefficient.** The average Spearman correlation in the annual attack rate between epidemic years (as a function of the number of years between epidemics), estimated over the epidemic years 20–159 years after the pandemic. Estimates are shown with their mean (points) and 95% confidence interval (bars) for different durations of short-term immunity (colours, see legend). The dashed black line highlights a correlation of 0 (no correlation). (TIFF)

S1 Table. Average annual attack rates for each decade, for different values of the antigenic seniority parameter. (XLSX)

S2 Table. All parameter values used in the model for the main analysis and the parameter values used (one parameter value varied at a time) in sensitivity analyses.

(XLSX)

**S3 Table.** Contact rates between all age-groups. (XLSX)

#### **Author contributions**

Conceptualization: Oliver Eales, Freya M. Shearer, James M. McCaw.

Formal analysis: Oliver Eales.

Methodology: Oliver Eales.

Software: Oliver Eales.

Supervision: Freya M. Shearer, James M. McCaw.

Visualization: Oliver Eales.

Writing – original draft: Oliver Eales.

Writing - review & editing: Oliver Eales, Freya M. Shearer, James M. McCaw.

#### References

- Bedford T, Suchard MA, Lemey P, Dudas G, Gregory V, Hay AJ, et al. Integrating influenza antigenic dynamics with molecular evolution. Elife. 2014;3:e01914. <u>https://doi.org/10.7554/eLife.01914</u> PMID: 24497547
- Medina RA, García-Sastre A. Influenza A viruses: new research developments. Nat Rev Microbiol. 2011;9(8):590–603. https://doi.org/10.1038/nrmicro2613 PMID: 21747392
- Nelson MI, Simonsen L, Viboud C, Miller MA, Holmes EC. Phylogenetic analysis reveals the global migration of seasonal influenza A viruses. PLoS Pathog. 2007;3(9):1220–8. <u>https://doi.org/10.1371/</u> journal.ppat.0030131 PMID: <u>17941707</u>
- Viboud C, Alonso WJ, Simonsen L. Influenza in tropical regions. PLoS Med. 2006;3(4):e89. <u>https://doi.org/10.1371/journal.pmed.0030089</u> PMID: 16509764
- Bedford T, Riley S, Barr IG, Broor S, Chadha M, Cox NJ, et al. Global circulation patterns of seasonal influenza viruses vary with antigenic drift. Nature. 2015;523(7559):217–20. <u>https://doi.org/10.1038/</u> <u>nature14460</u> PMID: 26053121
- Russell CA, Jones TC, Barr IG, Cox NJ, Garten RJ, Gregory V, et al. The global circulation of seasonal influenza A (H3N2) viruses. Science. 2008;320(5874):340–6. <u>https://doi.org/10.1126/science.1154137</u> PMID: 18420927
- Smith DJ, Lapedes AS, de Jong JC, Bestebroer TM, Rimmelzwaan GF, Osterhaus ADME, et al. Mapping the antigenic and genetic evolution of influenza virus. Science. 2004;305(5682):371–6. <u>https://</u> doi.org/10.1126/science.1097211 PMID: 15218094
- Coudeville L, Bailleux F, Riche B, Megas F, Andre P, Ecochard R. Relationship between haemagglutination-inhibiting antibody titres and clinical protection against influenza: development and application of a bayesian random-effects model. BMC Medical Research Methodology. 2010;10:18. https://doi.org/10.1186/1471-2288-10-18
- 9. Kucharski AJ, Lessler J, Cummings DAT, Riley S. Timescales of influenza A/H3N2 antibody dynamics. PLoS Biol. 2018;16(8):e2004974. https://doi.org/10.1371/journal.pbio.2004974 PMID: 30125272
- Lessler J, Riley S, Read JM, Wang S, Zhu H, Smith GJD, et al. Evidence for antigenic seniority in influenza A (H3N2) antibody responses in southern China. PLoS Pathog. 2012;8(7):e1002802. <u>https:// doi.org/10.1371/journal.ppat.1002802</u> PMID: <u>22829765</u>
- Wen F, Bedford T, Cobey S. Explaining the geographical origins of seasonal influenza A (H3N2). Proc Biol Sci. 2016;283(1838):20161312. https://doi.org/10.1098/rspb.2016.1312 PMID: 27629034
- Ferguson NM, Galvani AP, Bush RM. Ecological and immunological determinants of influenza evolution. Nature. 2003;422(6930):428–33. <u>https://doi.org/10.1038/nature01509</u> PMID: <u>12660783</u>
- Koelle K, Cobey S, Grenfell B, Pascual M. Epochal evolution shapes the phylodynamics of interpandemic influenza A (H3N2) in humans. Science. 2006;314(5807):1898–903. <u>https://doi.org/10.1126/</u> science.1132745 PMID: 17185596
- Kucharski AJ, Lessler J, Read JM, Zhu H, Jiang CQ, Guan Y, et al. Estimating the life course of influenza A(H3N2) antibody responses from cross-sectional data. PLoS Biol. 2015;13(3):e1002082. https://doi.org/10.1371/journal.pbio.1002082 PMID: 25734701
- 15. Bedford T, Rambaut A, Pascual M. Canalization of the evolutionary trajectory of the human influenza virus. BMC Biol. 2012;10:38. https://doi.org/10.1186/1741-7007-10-38 PMID: 22546494
- Peppa M, John Edmunds W, Funk S. Disease severity determines health-seeking behaviour amongst individuals with influenza-like illness in an internet-based cohort. BMC Infect Dis. 2017;17(1):238. https://doi.org/10.1186/s12879-017-2337-5 PMID: 28359335
- Eales O, McCaw JM, Shearer FM. Biases in routine influenza surveillance indicators used to monitor infection incidence and recommendations for improvement. Influenza Other Respi Viruses. 2024;18:e70050. https://onlinelibrary.wiley.com/doi/10.1111/irv.70050
- Eales O, McCaw JM, Shearer FM. Challenges in the case-based surveillance of infectious diseases. R Soc Open Sci. 2024;11(8):240202. <u>https://doi.org/10.1098/rsos.240202</u> PMID: <u>39205993</u>

- Archived: Estimated Influenza Illnesses, Medical visits, Hospitalizations, and Deaths in the United States — 2019–2020 Influenza Season. 10 May 2023 [cited 20 Jun 2023]. Available from: <u>https://www. cdc.gov/flu/about/burden/2019-2020/archive-09292021.html</u>
- 20. World Population Prospects Population Division United Nations. [cited 8 Sep 2023]. Available from: https://population.un.org/wpp/Graphs/DemographicProfiles/Pyramid/900
- Czaja CA, Miller L, Alden N, Wald HL, Cummings CN, Rolfes MA, et al. Age-Related Differences in Hospitalization Rates, Clinical Presentation, and Outcomes Among Older Adults Hospitalized With Influenza-U.S. Influenza Hospitalization Surveillance Network (FluSurv-NET). Open Forum Infect Dis. 2019;6(7):ofz225. <u>https://doi.org/10.1093/ofid/ofz225</u> PMID: <u>31363771</u>
- Geard N, Glass K, McCaw JM, McBryde ES, Korb KB, Keeling MJ, et al. The effects of demographic change on disease transmission and vaccine impact in a household structured population. Epidemics. 2015;13:56–64. https://doi.org/10.1016/j.epidem.2015.08.002 PMID: 26616042
- Feehan DM, Mahmud AS. Quantifying population contact patterns in the United States during the COVID-19 pandemic. Nat Commun. 2021;12(1):893. <u>https://doi.org/10.1038/s41467-021-20990-2</u> PMID: <u>33563992</u>
- Dhanasekaran V, Sullivan S, Edwards KM, Xie R, Khvorov A, Valkenburg SA, et al. Human seasonal influenza under COVID-19 and the potential consequences of influenza lineage elimination. Nat Commun. 2022;13(1):1721. <u>https://doi.org/10.1038/s41467-022-29402-5</u> PMID: <u>35361789</u>
- Chan KF, Carolan LA, Korenkov D, Druce J, McCaw J, Reading PC, et al. Investigating Viral Interference Between Influenza A Virus and Human Respiratory Syncytial Virus in a Ferret Model of Infection. J Infect Dis. 2018;218(3):406–17. https://doi.org/10.1093/infdis/jiy184 PMID: 29746640
- Laurie KL, Guarnaccia TA, Carolan LA, Yan AWC, Aban M, Petrie S, et al. Interval Between Infections and Viral Hierarchy Are Determinants of Viral Interference Following Influenza Virus Infection in a Ferret Model. J Infect Dis. 2015;212(11):1701–10. <u>https://doi.org/10.1093/infdis/jiv260</u> PMID: 25943206
- Vinh DN, Nhat NTD, de Bruin E, Vy NHT, Thao TTN, Phuong HT, et al. Age-seroprevalence curves for the multi-strain structure of influenza A virus. Nat Commun. 2021;12(1):6680. <u>https://doi.org/10.1038/</u> s41467-021-26948-8 PMID: 34795239
- Somes MP, Turner RM, Dwyer LJ, Newall AT. Estimating the annual attack rate of seasonal influenza among unvaccinated individuals: A systematic review and meta-analysis. Vaccine. 2018;36(23):3199– 207. https://doi.org/10.1016/j.vaccine.2018.04.063 PMID: 29716771
- Yang W, Lipsitch M, Shaman J. Inference of seasonal and pandemic influenza transmission dynamics. Proc Natl Acad Sci U S A. 2015;112(9):2723–8. <u>https://doi.org/10.1073/pnas.1415012112</u> PMID: 25730851
- Hayward AC, Fragaszy EB, Bermingham A, Wang L, Copas A, Edmunds WJ, et al. Comparative community burden and severity of seasonal and pandemic influenza: results of the Flu Watch cohort study. Lancet Respir Med. 2014;2(6):445–54. <u>https://doi.org/10.1016/S2213-2600(14)70034-7</u> PMID: 24717637
- 31. Huang QS, Bandaranayake D, Wood T, Newbern EC, Seeds R, Ralston J, et al. Risk Factors and Attack Rates of Seasonal Influenza Infection: Results of the Southern Hemisphere Influenza and Vaccine Effectiveness Research and Surveillance (SHIVERS) Seroepidemiologic Cohort Study. J Infect Dis. 2019;219(3):347–57. https://doi.org/10.1093/infdis/jiy443 PMID: 30016464
- Cauchemez S, Horby P, Fox A, Mai LQ, Thanh LT, Thai PQ, et al. Influenza infection rates, measurement errors and the interpretation of paired serology. PLoS Pathog. 2012;8(12):e1003061. <u>https://doi.org/10.1371/journal.ppat.1003061</u> PMID: 23271967
- Wen FT, Malani A, Cobey S. The Potential Beneficial Effects of Vaccination on Antigenically Evolving Pathogens. Am Nat. 2022;199(2):223–37. <u>https://doi.org/10.1086/717410</u> PMID: <u>35077280</u>
- Rouzine I, Rozhnova G. Antigenic evolution of viruses in host populations. PLoS Pathogens. 2018;14:e1007291. https://doi.org/10.1371/journal.ppat.1007291
- Rota PA, Wallis TR, Harmon MW, Rota JS, Kendal AP, Nerome K. Cocirculation of two distinct evolutionary lineages of influenza type B virus since 1983. Virology. 1990;175(1):59–68. <u>https://doi.org/10.1016/0042-6822(90)90186-u</u> PMID: 2309452
- Nextstrain / seasonal-flu / h3n2 / ha / 12y@2024-05-27. In: Nexstrain [Internet]. [cited 4 Oct 2024]. Available from: https://nextstrain.org/seasonal-flu/h3n2/ha/12y@2024-05-27
- Morris D, Gostic K, Pompei S, Bedford T, Łuksza M, Neher R, et al. Predictive modeling of influenza shows the promise of applied evolutionary biology. Trends in Microbiology. 2018;26:102–18.
- Palache A, Oriol-Mathieu V, Abelin A, Music T. Seasonal influenza vaccine dose distribution in 157 countries (2004-2011). Vaccine. 2014;32:6369–76.

- Pan K, Deem M. Comment on Ndifon *et al.*, "On the use of hemagglutination-inhibition for influenza surveillance: surveillance data are predictive of influenza vaccine effectiveness." Vaccine. 2009;27:5033–4.
- 40. Nordin J, Mullooly J, Poblete S, Strikas R, Petrucci R, Wei F, et al. Influenza vaccine effectiveness in preventing hospitalizations and deaths in persons 65 years or older in Minnesota, New York, and Oregon: data from 3 health plans. J Infect Dis. 2001;184(6):665–70. <u>https://doi.org/10.1086/323085</u> PMID: <u>11517426</u>
- **41.** Medlock J, Galvani AP. Optimizing influenza vaccine distribution. Science. 2009;325(5948):1705–8. https://doi.org/10.1126/science.1175570 PMID: <u>19696313</u>
- Worby CJ, Chaves SS, Wallinga J, Lipsitch M, Finelli L, Goldstein E. On the relative role of different age groups in influenza epidemics. Epidemics. 2015;13:10–6. <u>https://doi.org/10.1016/j.epidem.2015.04.003</u> PMID: 26097505
- Ainslie KEC, Haber M, Orenstein WA. Challenges in estimating influenza vaccine effectiveness. Expert Rev Vaccines. 2019;18(6):615–28. <u>https://doi.org/10.1080/14760584.2019.1622419</u> PMID: <u>31116070</u>
- 44. Fox A, Carolan L, Leung V, Phuong HVM, Khvorov A, Auladell M, et al. Opposing Effects of Prior Infection versus Prior Vaccination on Vaccine Immunogenicity against Influenza A(H3N2) Viruses. Viruses. 2022;14(3):470. <u>https://doi.org/10.3390/v14030470</u> PMID: <u>35336877</u>
- Ranjeva S, Subramanian R, Fang V, Leung G, Ip D, Perera R. Age-specific differences in the dynamics of protective immunity to influenza. Nature Communications. 2019;10(1):1660. <u>https://doi.org/10.1038/s41467-019-09645-0</u>
- 46. Pica N, Hai R, Krammer F, Wang TT, Maamary J, Eggink D, et al. Hemagglutinin stalk antibodies elicited by the 2009 pandemic influenza virus as a mechanism for the extinction of seasonal H1N1 viruses. Proc Natl Acad Sci U S A. 2012;109(7):2573–8. <u>https://doi.org/10.1073/pnas.1200039109</u> PMID: <u>22308500</u>
- 47. Skowronski D, Hottes T, McElhaney J, Janjua N, Sabaiduc S, Chan T, et al. Immuno-epidemiologic correlates of pandemic H1N1 surveillance observations: higher antibody and lower cell-mediated immune responses with advanced age. J Infect Dis. 2011;203:158–67.
- Mykytyn AZ, Rissmann M, Kok A, Rosu ME, Schipper D, Breugem TI, et al. Antigenic cartography of SARS-CoV-2 reveals that Omicron BA.1 and BA.2 are antigenically distinct. Sci Immunol. 2022;7:eabq4450. https://doi.org/10.1126/sciimmunol.abq4450
- Hornsby H, Nicols AR, Longet S, Liu C, Tomic A, Angyal A, et al. Omicron infection following vaccination enhances a broad spectrum of immune responses dependent on infection history. Nat Commun. 2023;14(1):5065. https://doi.org/10.1038/s41467-023-40592-4 PMID: 37604803
- Reynolds CJ, Pade C, Gibbons JM, Otter AD, Lin K-M, Muñoz Sandoval D, et al. Immune boosting by B.1.1.529 (Omicron) depends on previous SARS-CoV-2 exposure. Science. 2022;377(6603):eabq1841. <u>https://doi.org/10.1126/science.abq1841</u> PMID: <u>35699621</u>
- Eales O, Plank MJ, Cowling BJ, Howden BP, Kucharski AJ, Sullivan SG, et al. Key Challenges for Respiratory Virus Surveillance while Transitioning out of Acute Phase of COVID-19 Pandemic. Emerg Infect Dis. 2024;30(2):e230768. <u>https://doi.org/10.3201/eid3002.230768</u> PMID: <u>38190760</u>
- 52. Hoffman M, Gelman A. The No-U-Turn Sampler: Adaptively Setting Path Lengths in Hamiltonian Monte Carlo. arXiv [stat.CO]. 2011.
- Nakamura GM, Martinez AS. Hamiltonian dynamics of the SIS epidemic model with stochastic fluctuations. Sci Rep. 2019;9(1):15841. <u>https://doi.org/10.1038/s41598-019-52351-x</u> PMID: <u>31676857</u>
- 54. Tierney N, Golding N, Babu A. conmat: Builds contact matrices using GAMs and population data. 2022.
- Mossong J, Hens N, Jit M, Beutels P, Auranen K, Mikolajczyk R, et al. Social contacts and mixing patterns relevant to the spread of infectious diseases. PLoS Med. 2008;5(3):e74. <u>https://doi.org/10.1371/</u> journal.pmed.0050074 PMID: <u>18366252</u>
- 56. Tsai M-T, Chern T-C, Chuang J-H, Hsueh C-W, Kuo H-S, Liau C-J, et al. Efficient simulation of the spatial transmission dynamics of influenza. PLoS One. 2010;5(11):e13292. <u>https://doi.org/10.1371/journal.pone.0013292</u> PMID: 21079810
- Wood SN. Thin plate regression splines. J Royal Stat Soc Series B (Statistical Methodology). 2003;65:95–114.
- Wenger JB, Naumova EN. Seasonal synchronization of influenza in the United States older adult population. PLoS One. 2010;5(4):e10187. https://doi.org/10.1371/journal.pone.0010187 PMID: 20419169